Author's response to reviews

Title: Estimating Multimorbidity Disease Burden Among Older Adults: A Study of Six Chronic Illness Measures, 2008/09 Canadian Community Health Survey

Authors:

Andrew V. Wister (wister@sfu.ca)
Mélanie Levasseur (Melanie.Levasseur@USherbrooke.ca)
Lauren Griffith (griffith@mcmaster.ca)
Ian Fyffe (iaf@sfu.ca)

Version: 2
Date: 1 August 2014

Author's response to reviews:

Added:

1) There are no competing interests, and this has been included.
2) Authors’ contributions have been detailed under that subtitle.
3) There are no acknowledgements.
4) Line numbering has been added.
5. Ethics details have been added under that subtitle.